Author’s response to reviews

Title: Posterior Short Segment Pedicle Screw Fixation and TLIF for the Treatment of Unstable Thoracolumbar Fracture

Authors:

Ling Wang (whwl@hotmail.com)
Jianjun Li (lijianjun71@yahoo.com.cn)
Hong Wang (whwl@hotmail.com)
Qun Yang (yangqun@medmail.com.cn)
Decheng Lv (lvdc@dmu-1.com)
Weiguo Zhang (zwg@medmail.com.cn)
Xianping Shang (shangxp1989@126.com)
Depeng Kou (koudepeng@126.com)
Xunyuan Jia (805343581@qq.com)
Xianglong Yang (yangxlcq@126.com)
Yilong Tang (tang.yilong@163.com)
Meng Zhang (105009599@qq.com)
Pengrui Wang (374576889@qq.com)
Yan Xu (522566593@qq.com)
Shijin Wang (2298022330@qq.com)

Version: 2 Date: 12 August 2013

Author’s response to reviews:

1. All the 5 kindly required sections were added, including "Competing interests, Authors' contributions, Acknowledgements, Ethics statement and Consent statement".

2. Thank you for your kind information, text from Figure 5 F and L were completely removed.

3. We clarify that we have obtained consent from patients for publication of their individual details on the premise of never divulging a real name.